Author’s response to reviews

Title: Implementation and Dissemination of a Transition of Care Program for Rural Veterans: A Controlled Before and After Study

Authors:

Chelsea Leonard (leonard.chelsea@gmail.com)
Emily Lawrence (Emily.Lawrence2@va.gov)
Marina McCreight (Marina.Mccreight@va.gov)
Brandi Lippman (Brandi.Lippmann@va.gov)
Lynette Kelley (Lynette.Kelley@va.gov)
Ashlea Mayberry (Ashlea.Mayberry@va.gov)
Amy Ladebue (Amy.Ladebue@va.gov)
Heather Gilmarin (heather.gilmartin@va.gov)
Murray Côté (mjcote@sph.tamhsc.edu)
Jacqueline Jones (Jacqueline.Jones@ucdenver.edu)
Borsika Rabin (borsika.a.rabin@gmail.com)
Michael Ho (Michael.Ho@va.gov)
Robert Burke (Robert.Burke5@va.gov)

Version: 2 Date: 28 Sep 2017

Author’s response to reviews:

9/28/2017
Dear Editors,

Thank you for your comments on our manuscript. We have addressed your suggestions in the following ways. 1) We changed the title to “Implementation and Dissemination of a Transition of Care Program for Rural Veterans: A Controlled Before and After Study.” 2) We added a brief
description of our outcomes measures to the abstract. 3) We reformatted the methods section to follow the StaRI reporting format. We begin this section with a description of the intervention to improve flow, and then describe the study design, context, and implementation strategy, followed by a methodological overview of our intervention and implementation outcomes. These changes are on pages 6 to 13 of the manuscript. 4) We replaced the PRISM figure with a flow chart describing our study. Because we felt that this made Figure 2 redundant, we omitted Figure 2.

We feel that these changes improve the manuscript. Thank you for your consideration.

Sincerely,

Chelsea Leonard